Author’s response to reviews

Title: Evaluation of Pharmacological Activities and Assessment of Intraocular Penetration of an Ayurvedic Polyherbal Eye Drop (ItoneTM) in Experimental Models

Authors:

Thirumurthy Velpandian (tvelpandian@hotmail.com)
Pankaj Gupta (gupta.aiims@gmail.com)
Alok Kumar Ravi (ravialok@yahoo.com)
Hanuman Prasad Sharma (coolhanusharma@gmail.com)
Nihar Ranjan Biswas (nr_biswas@yahoo.com)

Version: 4 Date: 10 December 2012

Author’s response to reviews: see over
To, The Editor in Chief
The Editor in Chief
BMC Complementary & Alternative Medicine

Sub: Submission of revised manuscript of ID 1620577892776062

Dear Sir / Madam,

We thank you for your valuable suggestion in our manuscript. We have accepted it and appropriate correction has been made in the competing interest section. The changes made are enclosed in the following page. I hope you may now find it suitable for the publication.

We once again thank you for your cooperation.

Thanking you

Yours sincerely

Dr. T. Velpandian
Corresponding author
Editor’s suggestion
Thank you for clarifying the role of Dey's Medical Store in your study. We understand that they played no role in the design, implementation or analysis of your study. However, we would consider the fact that they provided financial assistance to your study to constitute a potential competing interest, which could be misconstrued by readers of your paper. We would therefore ask you to revise your competing interests section to clearly declare the role of Dey's Medical Store in your study, and stating that they are the manufacturers of Itone. Please also declare that they had no role in the design, implementation or interpretation of your study.

Revised Competing Interest section
Dey’s Medical Store, Kolkata is the manufacturer of Itone™ who has provided preservative free formulation and academic research grant for conducting this scientific study. The authors hereby declare that the manufacturer had no role or interest in the design, conduct and interpretation of the study.